%&tex
\chapter{Analysis}
\label{analysis}
The goal of this chapter is to interpret the design method and crystallize it into a design plan.
\autoref{sec:design_flow} introduces a design flow that contains the basic design steps.
These steps are still undefined but are materialized with the techniques from the design method in \autoref{sec:design_refinement}.
The remaining steps that are not covered by the design method are defined in \autoref{sec:design_extension}.

\section{Design Flow}
\label{sec:design_flow}
A waterfall based design approach is used as the basis for the \emph{design flow}.
The most basic form for a waterfall is shown in \autoref{fig:waterfall}.
The implementation and testing steps will be 

The design method by \textcite{broenink_rapid_2019} consists of three main cycles: Preparatory work, Inner Cycle, and Outer Cycle.
These steps describe the direct 
The aim of the preparatory work is to split and order the features and the levels of detail.




\begin{marginfigure}
    \centering
    \includegraphics[width=2.9cm]{graphics/waterfall.pdf}
    \caption{Minimalistic implementation of waterfall model.}
    \label{fig:waterfall}
\end{marginfigure}









\section{Refinement}
\label{sec:design_refinement}


\section{Extension}
\label{sec:design_extension}

